Author’s response to reviews

Title: Factors associated with infants' and young children's (6-23 months) dietary diversity in Pakistan: Evidence from the Demographic and Health Survey 2012-13

Authors:

Sarosh Iqbal (sarosh.iqbal@gmail.com)
Rubeena Zakar (rubeena499@hotmail.com)
Muhammad Zakar (mzzakir@yahoo.com)
Florian Fischer (f.fischer@uni-bielefeld.de)

Version: 5 Date: 15 Nov 2017

Author’s response to reviews:

Please find attached the manuscript with the corrections in Table 1.